Author's response to reviews

Title: A community-based pilot randomised controlled study of life skills classes for individuals with low mood and depression.

Authors:

Carrie-Anne McClay (c.mcclay.1@research.gla.ac.uk)
Katrina Collins (drkcollins@btconnect.com)
Lynsay Matthews (Lynsay.Matthews@uws.ac.uk)
Caroline Haig (caroline.haig@glasgow.ac.uk)
Alex McConnachie (alex.mcconnachie@glasgow.ac.uk)
Jill Morrison (jill.morrison@glasgow.ac.uk)
Pat Lynch (pat@aware-ni.org)
Louise Waters (louisemewan@live.co.uk)
Ilena Day (ilenad@actionondepression.org)
Grainne McAnee (McAnee-G@email.ulster.ac.uk)
Chris Williams (chris.williams@glasgow.ac.uk)

Version: 5
Date: 18 December 2014

Author's response to reviews: see over
Dear Dr van Ballegooijen,

The research team are delighted that the above manuscript has been accepted in principle.

The trial registration information in the abstract has been amended as requested.

Yours sincerely,

Professor Chris Williams
Professor of Psychosocial Psychiatry and Honorary Consultant Psychiatrist